Title: Do electronic health records affect the patient-psychiatrist relationship? A before & after study of psychiatric outpatients

Authors:

Randall F. Stewart (randallfstewart@gmail.com)
Robert Bailey (bbailey@salud.unm.edu)
Mark Schuyler (mschuyler@salud.unm.edu)
Philip J. Kroth (pkroth@salud.unm.edu)

Version: 8 Date: 6 January 2010

Author's response to reviews:

Minor edits to:

1. Be more consistent in Philip J. Kroth's initials. All "PK" changed to "PJK." The initials "PKR" that are next to the e-mail address on p. 2 were corrected to "PJK."

2. Order of author's names finalized to show strict order of involvement instead of the original approach which was to list the primary mentor last.

3. Minor artistic changes to remove unnecessary commas and simplify 2 or 3 phrases.